Author's response to reviews

Title: A qualitative study on barriers of managing anaemia during pregnancy in Public Health Centres: perceptions of Indonesian nurse-midwives.

Authors:

Widyawati - Widyawati - (a.widyawati@radboudumc.nl)
Suze - Jans - (suzejans@dds.nl)
Sutarti - Utomo - (heraan@gmail.com)
Jeroen van Dillen - (jeroen.vandillen1@radboudumc.nl)
Toine Lagro Janssen - (toine.lagro@radboudumc.nl)

Version: 4
Date: 25 April 2014

Author's response to reviews:

Details of revisions:

1. line numbering and page number on the main manuscript.

2. do not insert page break in the manuscript